Author's response to reviews

Title: The impact of the facial skeletal class on the psychological profile: an epidemiological survey on young Italian adults.

Authors:

Andrea Senna (andreasenna75@hotmail.com)
Daniele Abbenante (dabbenante@hotmail.it)
Lucio Tremolizzo (lucio.tremolizzo@unimib.it)
Guglielmo Campus (gcampus@uniss.it)
Laura Strohmenger (laura.strohmenger@unimi.it)

Version: Date: 2 April 2006

Author's response to reviews:

To Miss Dr Jo Appleford
Assistant Editor,

Dear Miss Appleford, I modified the paper following your suggestions:
1-a title page (with names of the authors, institutions, countries and email addresses of all authors, and the full postal address of the submitting author) is included;
2-the ethical permission is now included in Material and Methods sections;
3-the term NS is now modified in not-S (not-successful)

Thanks in advance for your kind support
Yours
Guglielmo Campus